\documentclass[12pt]{article}
\usepackage{arial,url}

\begin{document}

\title{A LaTeX document that uses citations}
\author{Linda Cecilie Kjeldsen\\
        Faculty of Computer Sciences,\\
        Ostfold University College,\\
        Halden,\\
        Norway\\
    }
\date{\today}
\maketitle

\section*{}
This document is created partly to get to know \LaTeX\ , and partly to test the functions of BibTeX.
Following is a list over the subjects I will adress in my individual curriculum:
\begin{itemize}
\item Complete the implementation and testing of the program from the DigMap course
\item Implement and test a BLG-tree~\cite{blg}
\item Learn how to use \LaTeX\
\item Write an article
\item Implement and test a Reactive-Tree~\cite{reacttree} and a GAP-Tree~\cite{gap}
\end{itemize}

\subsection{More information}

At the present time, the first two points are in the finishing stage, and this document is a part of
the LaTeX-section of my curriculum. I will start writing the article very soon, and the final
implementation-point is optional, and will be omitted if time runs out...


\section{Citation section}
This section is really just to make sure all the entries in my bibliography-file are citated, so that
I can make sure that they are entered correctly. First of all I would like to refer to the article that
started this whole mess: The "Raster-Vector article" by Li and Openshaw~\cite{rastvect}. Then comes the
only book that I have bought, and it is about Computational Geometry~\cite{compgeom}. In books that I have
borrowed from my teaching supervisor, Gunnar Misund~\cite{gumi}, I have read the following articles:
\begin{itemize}
\item Generalization of Spatial Data: Principles and Selected Algorithms~\cite{general}
\item The Binary Line Generalization Tree~\cite{blg}
\item The Ractive-Tree~\cite{reacttree}
\end{itemize}
I will also read an article previously published by the journal~\cite{delaunay}. An important structure in
my previous work, is the priority search tree, presented by McCreight~\cite{pst_mccreight}, and described by
Prof. Tamassia~\cite{pst}. The other aspect of my work, in addition to generalization, is orthogonal range
queries, here presented by Piotr Indyk~\cite{ort}.

\begin{center}
\textit{Have a nice day :)}
\end{center}

\bibliographystyle{plain}
\bibliography{myBibFile}

\end{document}
